Author’s response to reviews

Title: Parkinsonism Secondary to Metastatic Lesions within the Central Nervous System: a Case Report

Authors:

Eduardo Hortelano (edhortel@hotmail.com)
Christian Perea (cperea@gmail.com)
Esther Uña (aunacid@hotmail.com)
Amelia Cebayos (ameliac@gmail.com)
Patricia Diezhandino (patridiezhandino@gmail.com)
Montserrat Gonzalez (montgonza@yahoo.es)

Version: 4 Date: 10 January 2010

Author’s response to reviews: see over
9th of January, 2010

Dear sir,

I have corrected the manuscript entitled “Central Nervous System Metastatic Lesions as a cause of Parkinsonism: a Case Report” as you have suggested.

RECOMMENDATIONS:

1. The term "CT" has been defined as "computed tomography" in the first use.
2. P. 6 1st paragraph the term "LDOPA" should be "L-DOPA". This has been corrected.
3. Use consistent spelling for the word "edema"--i.e., it is spelled "edema" p.6 2nd paragraph but also spelled "oedema" on p.4 3rd paragraph. I have corrected this.
4. p.6 1st paragraph of the conclusion section please change the second sentence to "However, the presence of an advanced carcinoma with rapid progression of parkinsonian symptoms should lead one to suspect neoplastic parkinsonism."
   Corrected.
5. Please consider using arrows in the figures to point out the mass lesions. I have signaled with arrows the lesions.
6. Please could you provide a statement to confirm that the patient has given their informed consent for the case report to be published. This should be included after the conclusion. We recommend the following wording is used for the consent section: "Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal." This has been included.

Esther Uña Cidón, MD, PhD and Professor
Medical Oncology Department
Clinical University Hospital
Valladolid (Spain)